Author’s response to reviews

Title: Posterior pole asymmetry analysis and retinal nerve fibre layer thickness measurements in primary angle-closure suspect patients

Authors:

Yi Zha (171652838@qq.com)
Wei Huang (906844251@qq.com)
Jinfei Zhuang (zhuangjf30@126.com)
Jianqiu Cai (drcaijq@163.com)

Version: 8 Date: 05 Jan 2019

Author’s response to reviews:

Dear editors:

I have tried my best to amend the overlap, especially showed in screenshot. but most overlap you showed are professional terminology, for example, primary angle-closure disease, primary angle-closure suspect, primary angle closure, posterior pole asymmetry analysis and peripapillary retinal nerve fiber layer, which I could not change. please check my revised version again and help me check other overlap again. thank you very much.

Sincerely

Jianqiu Cai